

Author's response to reviews

Title: Establishing confidence in the output of qualitative research synthesis: the ConQual approach

Authors:

Zachary Munn (Zachary.Munn@adelaide.edu.au)

Kylie Porritt (kylie.porritt@adelaide.edu.au)

Craig Lockwood (craig.lockwood@adelaide.edu.au)

Edoardo Aromataris (edoardo.aromataris@adelaide.edu.au)

Alan Pearson (alan.pearson@adelaide.edu.au)

Version: 5 **Date:** 28 July 2014

Author's response to reviews:

Thank you for your comments. We have made some changes to both of the figures.